Author’s response to reviews

Title: Mu opioid receptor availability in people with psychiatric disorders who died by suicide: A case control study.

Authors:

Elizabeth Scarr (elscarr@unimelb.edu.au)
Tammie T Money (ttmoney@unimelb.edu.au)
Geoffrey Pavey (g.pavey@mhri.edu.au)
Jaclyn Neo (j.neo@mhri.edu.au)
Brian Dean (anddali@unimelb.edu.au)

Version: 4 Date: 20 August 2012

Author’s response to reviews: see over
20th August 2012

Jigisha Patel MRCP, PhD
Series Editor – BMC Psychiatry
BioMed Central Ltd
Floor 6, 236 Gray's Inn Road
London
WC1X 8HB
United Kingdom

Dear Dr Patel:

Thank you for altering us to the issue with Table S1 in our manuscript “Mu opioid receptor availability in people with psychiatric disorders who died by suicide.” We do not have explicit permission from each individual or their next of kin to publish individual details. Therefore the table has been revised to contain only a summary of the information for each group. We trust that this will meet your requirements.

Thank you for your attention to this matter, we look forward to hearing from you.

Sincerely,

Elizabeth Scarr
ARC Future Fellow